Author’s response to reviews

Title: The effect of diluting povidone-iodine on bacterial growth associated with speech

Authors:

Sivashanth Gnanasekaran (gnanasekaran.sivashanth@gmail.com)
Sophie Rogers (srogers@unimelb.edu.au)
Sanj Wickremasinghe (Sanj.Wickremasinghe@eyeandear.org.au)
Sukhpal Sandhu (sukhpal.sandhu@unimelb.edu.au)

Version: 4 Date: 27 Jan 2019

Author’s response to reviews:

1. We notice that you have presented your abstract at the The Royal Australian and New Zealand College of Ophthalmologists, 47th Annual Scientific Congress, 1–4 November 2015:


Please ensure that this is cited and acknowledged in the manuscript.
We have cited and acknowledged this on the title page.

2. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

A statement has been included.

3. In the section 'Funding', please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

A statement has been added.